Author's response to reviews

Title: Short GSM mobile phone exposure does not alter human auditory brainstem response

Authors:

Gabor Stefanics (gstefan@cogpsyphy.hu)
Lorand Kellenyi (kellenyi@gamma.ttk.pte.hu)
Ferenc Molnar (molnar@hp.osski.hu)
Gyorgyi Kubinyi (bakos@hp.osski.hu)
Gyorgy Thuroczy (thuroczy@hp.osski.hu)
Istvan Hernadi (hernadi@ttk.pte.hu)

Version: 6 Date: 8 November 2007

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

Please find enclosed the revised version for manuscript (MS) entitled:

Stefanics G., et al.: Short GSM mobile phone exposure does not alter human auditory brain stem response. (MS: 4876010601424942)

Thank you for the formatting instructions to revise our manuscript. According to your instructions we have made the following changes in the manuscript:

We removed the figure and table placement text.

We indicated in the „Authors’ contributions” paragraph that all authors read and approved the final manuscript.

We rearranged the table titles and legend. The table titles now appear above the tables. The legends appear below the tables.

Best regards,

Istvan Hernadi